[role="xpack"]
[[apm-getting-started]]
== Get started with the APM app

++++
<titleabbrev>Get started</titleabbrev>
++++

// Conditionally display a screenshot or video depending on what the
// current documentation version is.

ifeval::["{is-current-version}"=="true"]
++++
<script type="text/javascript" async src="https://play.vidyard.com/embed/v4.js"></script>
<img
  style="width: 100%; margin: auto; display: block;"
  class="vidyard-player-embed"
  src="https://play.vidyard.com/Y4nE2XLYEk75odbRQmUA3g.jpg"
  data-uuid="Y4nE2XLYEk75odbRQmUA3g"
  data-v="4"
  data-type="inline"
/>
</br>
++++
endif::[]

For a quick, high-level overview of the health and performance of your application,
start with:

* <<services>>
* <<traces>>
* <<dependencies>>
* <<service-maps>>

Notice something awry? Select a service or trace and dive deeper with:

* <<service-overview>>
* <<mobile-service-overview>>
* <<transactions>>
* <<spans>>
* <<errors>>
* <<metrics>>
* <<infrastructure>>
* <<logs>>

TIP: Want to learn more about the Elastic APM ecosystem?
See the {apm-guide-ref}/apm-overview.html[APM Overview].

include::services.asciidoc[]

include::traces.asciidoc[]

include::dependencies.asciidoc[]

include::service-maps.asciidoc[]

include::service-overview.asciidoc[]

include::mobile-service.asciidoc[]

include::transactions.asciidoc[]

include::spans.asciidoc[]

include::errors.asciidoc[]

include::metrics.asciidoc[]

include::infrastructure.asciidoc[]

include::logs.asciidoc[]